Author's response to reviews

Title: Differences in cerebral response to esophageal acid stimuli and psychological anticipation in GERD subtypes - An fMRI study

Authors:

Kun Wang (wangkun55555@sina.com)
Li-Ping Duan (duanlp@bjmu.edu.cn)
Xiang-Zhu Zeng (rainman304@126.com)
Jian-Yu Liu (yueguang1997@126.com)
Xu-Chu Weng (wengxc@psych.ac.cn)

Version: 5 Date: 19 March 2011

Author's response to reviews:

Dear editor:

We have made some spelling correction in this revision as following:

1. In line 10, page 22, the "RE group, PFC: IC and ACC were activated." has been changed into "RE group: PFC, IC and ACC were activated."

2. In page 26, all "p" has been changed into capital "P".

3. In page 27, the "X, Y, Z" has been changed into "x, y, z".

4. In page 27, we have adjust the page margin of the table 3 in order to show all the data clearly.

Thank you very much!

Best regards!

Sincerely,

Li-Ping Duan